Author's response to reviews

Title: Essential trauma management: addressing service delivery needs in active conflict zones in eastern Myanmar

Authors:

Allison J Richard (sweetalmd@gmail.com)
Catherine I Lee (catherine@ghap.org)
Matthew G Richard (mattrichard@sbcglobal.net)
Eh Kalu Shwe Oo (ehkalushweoo@gmail.com)
Thomas Lee (tomlee@ucla.edu)
Lawrence Stock (laemdoc@aol.com)

Version: 5 Date: 25 February 2009

Author's response to reviews:

Revisions:

1) On authors page, names of two authors were changed. First, changed "Tom" to "Thomas" and second, changed "Larry" to "Lawrence"